

LSHTM Research Online

Harding-Esch, EM; Holland, MJ; Schémann, JF; Molina, S; Sarr, I; Andreasen, AA; Roberts, CH; Sillah, A; Sarr, B; Harding, EF; +3 more... Edwards, T; Bailey, RL; Mabey, DC; (2011) Diagnostic Accuracy of a Prototype Point-of-Care Test for Ocular Chlamydia trachomatis under Field Conditions in The Gambia and Senegal. PLoS neglected tropical diseases, 5 (8). e1234. ISSN 1935-2727 DOI: https://doi.org/10.1371/journal.pntd.0001234

Downloaded from: http://researchonline.lshtm.ac.uk/333/

 $DOI: \ https://doi.org/10.1371/journal.pntd.0001234$

Usage Guidlines:

Please refer to usage guidelines at http://researchonline.lshtm.ac.uk/policies.html or alternatively contact researchonline@lshtm.ac.uk.

Available under license: http://creativecommons.org/licenses/by/2.5/

Diagnostic Accuracy of a Prototype Point-of-Care Test for Ocular *Chlamydia trachomatis* under Field Conditions in The Gambia and Senegal

Emma M. Harding-Esch¹, Martin J. Holland^{1,2}, Jean-François Schémann³, Sandra Molina¹, Isatou Sarr², Aura A. Andreasen¹, Chrissy h. Roberts¹, Ansumana Sillah⁴, Boubacar Sarr⁵, Edward F. Harding⁶, Tansy Edwards¹, Robin L. Bailey¹, David C. W. Mabey¹*

1 London School of Hygiene and Tropical Medicine, London, United Kingdom, 2 Medical Research Council Laboratories, Fajara, Banjul, The Gambia, 3 Institut de Recherche pour le Développement (IRD), Dakar, Sénégal, 4 National Eye Care Programme, Ministry of Health, Banjul, The Gambia, 5 Programme National de Lutte contre la Cécité, Ministère de la Santé, Dakar, Sénégal, 6 Ely, United Kingdom

Abstract

Background: The clinical signs of active trachoma are often present in the absence of ocular *Chlamydia trachomatis* infection in low prevalence and mass treated settings. Treatment decisions are currently based on the prevalence of clinical signs, and this may result in the unnecessary distribution of mass antibiotic treatment. We aimed to evaluate the diagnostic accuracy of a prototype point-of-care (POC) test, developed for field diagnosis of ocular *C. trachomatis*, in low prevalence settings of The Gambia and Senegal.

Methodology/Principal Findings: Three studies were conducted, two in The Gambia and one in Senegal. Children under the age of 10 years were screened for the clinical signs of trachoma. Two ocular swabs were taken from the right eye. The first swab was tested by the POC test in the field and the result independently graded by two readers. The second swab was tested for the presence of *C. trachomatis* by Amplicor Polymerase Chain Reaction. In Senegal, measurements of humidity and temperature in the field were taken. A total of 3734 children were screened, 950 in the first and 1171 in the second Gambian study, and 1613 in Senegal. The sensitivity of the prototype POC test ranged between 33.3–67.9%, the specificity between 92.4–99.0%, the positive predictive value between 4.3–21.0%, and the negative predictive value between 98.0–99.8%. The rate of false-positives increased markedly at temperatures above 31.4°C and relative humidities below 11.4%.

Conclusions/Significance: In its present format, this prototype POC test is not suitable for field diagnosis of ocular *C. trachomatis* as its specificity decreases in hot and dry conditions: the environment in which trachoma is predominantly found. In the absence of a suitable test for infection, trachoma diagnosis remains dependent on clinical signs. Under current WHO recommendations, this is likely resulting in the continued mass treatment of non-infected communities.

Citation: Harding-Esch EM, Holland MJ, Schémann J-F, Molina S, Sarr I, et al. (2011) Diagnostic Accuracy of a Prototype Point-of-Care Test for Ocular Chlamydia trachomatis under Field Conditions in The Gambia and Senegal. PLoS Negl Trop Dis 5(8): e1234. doi:10.1371/journal.pntd.0001234

Editor: Julius Schachter, University of California San Francisco, United States of America

Received October 3, 2010; Accepted May 22, 2011; Published August 2, 2011

Copyright: © 2011 Harding-Esch et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The authors are grateful to the Wellcome Trust for funding this study (grant number 078460/Z/05/Z). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: david.mabey@lshtm.ac.uk

Introduction

Trachoma is caused by ocular infection with the bacterium *Chlamydia trachomatis* and is the leading infectious cause of blindness worldwide [1]. The World Health Organization (WHO) simplified grading system, designed for the simple and reliable grading of trachoma clinical signs by non-specialist staff, is predominantly used for trachoma diagnosis in the field [2]. This system classifies the clinical signs into five categories: trachomatous inflammation-follicular (TF), trachomatous inflammation-intense (TI), trachomatous scarring (TS), trachomatous trichiasis (TT), and corneal opacity (CO).

Clinical signs are however poorly correlated with detection of ocular *C. trachomatis* infection, since they may persist for months or years after infection has cleared [3,4,5,6,7]. The WHO recommends

that any district or community where the prevalence of TF in children aged 1–9 years is at least 10% should receive mass antibiotic treatment annually for three years, before the prevalence is reassessed [8]. Since antibiotics are given to treat *C. trachomatis* infection, and the prevalence of clinical signs is a poor predictor of infection especially in low prevalence and mass treated settings, treatment may be unnecessarily commenced and continued, thus wasting scarce resources. A point-of-care (POC) test capable of detecting infection in the field would enable treatment to be directed to those communities in need. Since a POC test would be used to make treatment decisions at the community, rather than the individual, level, it is important that it has high specificity (>98%), otherwise it has no advantage over the use of clinical signs.

A prototype POC test for trachoma, developed by the Diagnostics Development Unit (University of Cambridge, UK),

Author Summary

Trachoma, caused by infection of the eye with the bacterium Chlamydia trachomatis, is the leading infectious cause of blindness and is associated with poverty. Antibiotic treatment of all community members is one of the recommended control strategies for trachoma. However, in places where the prevalence of clinical signs is low, C. trachomatis eye infection is often absent. Laboratory testing for C. trachomatis infection by polymerase chain reaction (PCR) is highly sensitive but expensive and requires well-trained staff. A simple point-of-care (POC) test that can be used in trachoma-affected communities could help trachoma control efforts. We evaluated a POC test for C. trachomatis eye infection. Children under 10 years of age were screened for clinical signs of trachoma and C. trachomatis eye infection. The POC test result was compared with laboratory PCR test results. The POC test detected just over half of PCR test positives correctly. However, the POC test tended to give false-positive results in hot and dry conditions, which is the typical environment of trachoma. The POC test requires high specificity since it would be used to make treatment decisions at the community level. Therefore, its present format requires improvement before it can be utilized in trachoma control.

and currently not commercially available, has previously been evaluated on a small scale in a medium prevalence Tanzanian setting (12.5–37.9% TF in children aged 1–9 years), with encouraging results [9]. This assay is a modified version of a test for genital *C. trachomatis* infection [10,11], optimised for use with conjunctival swabs. The assay detects the chlamydial lipopolysaccharide (LPS), using lateral flow technology. The dipstick is made up of a nitrocellulose membrane affixed to a backing sheet, and connected to an absorbent pad, with two immobilised monoclonal antibodies (mAbs) lined on the dipstick membrane. The mAb at the capture line is against chlamydial LPS, and that at the procedural control line is an antibiotin antibody. This assay was designed specifically for use in resource-limited settings, and therefore has no electricity, water or laboratory equipment requirements [9].

We aimed to conduct a larger scale evaluation of this prototype POC test's diagnostic accuracy in children aged under 10 years in the low prevalence settings of The Gambia and Senegal. The functional temperature and humidity range of the prototype test was unknown before this study's field testing.

Methods

The study has been reported in accordance with the STARD (STAndards for the Reporting of Diagnostic accuracy studies) checklist (provided as Supporting Information S1) [12].

Ethics statement

Research was done in accordance with the declaration of Helsinki. Ethical approval was obtained from the London School of Hygiene & Tropical Medicine (LSHTM) ethics committee (No.2067), the Gambia Government/Medical Research Council Joint Ethics Committee (SCC 979), and the Comité d'éthique du CNRS, Dakar, Senegal. Written (thumbprint or signature) informed consent was obtained from the guardians of all children.

Study site and participant selection

Three studies were conducted, two in The Gambia and one in Senegal. An overview of the study methods is depicted in Figure 1. Study 1 was part of a survey of the Lower River (LRR) and North Bank (NBR) Regions of The Gambia. The sample selection has been described in detail elsewhere [13]. Briefly, 19 census Enumeration Areas (EAs), which are designed to be of approximately the same population size, were randomly selected in LRR. A random selection of households was made so that 50 children aged under 10 years would be included. In Studies 2 and 3, all children aged under 10 years were included. Study 2 took place in 6 Gambian communities and Study 3 in 12 Senegalese communities. The Gambian communities were selected on the basis of having a TF prevalence of at least 10% in the Gambian survey [13], increasing the likelihood of finding infection. Study 3 was based in the health post of Keur Samba Kane in Bambey District, which had been identified by the National Eve Care Programme as fulfilling the WHO criteria for mass treatment. Study 1 was conducted in January-March 2006, Study 2 in March-May 2006, and Study 3 in January-February 2007.

Census enumeration

The village head (*alkalo*) and villagers were sensitised to the study's aims and methods. Household head lists were made and the *de facto* population was enumerated, recording their name, alias names, age and sex. Date-of-birth was noted when possible using ID cards and infant vaccination cards. The census team identified eligible children and informed household heads of the day and place of examination to ensure optimum participation.

Clinical examination

Experienced Gambian and Senegalese graders were used. Their grading was verified and standardised using WHO grading slides, and a chance corrected agreement (Cohen's kappa [14]) score of at least 0.8 was required for the scoring of each sign (TF, TI, TS, TT). NBR villages with the highest active trachoma prevalence in the Gambian survey were re-visited and children diagnosed with active disease were re-screened by a senior grader to verify clinical diagnoses.

The examination team located itself in a central point in the village. Eligible children were called and written informed consent (signature or thumbprint) from the participants' guardians was obtained. The validated grader examined each consenting participant's eyes using a $2.5 \times$ magnifying loupe and torchlight. In order to avoid cross-contamination, the examiner wore and changed gloves between each participant. The clinical diagnosis was made according to the WHO simplified grading system [2].

Ocular sample collection

Two swabs were taken from the tarsal conjunctiva of each participant's right eye using a standardised technique [15], whereby the swab was held horizontally and drawn lengthways across the everted upper tarsal conjunctiva four times, rotating the head of the swab a quarter turn with each pass. Crosscontamination of samples was limited by using a field worker to pass the swab to the examiner. The field worker then held the tube into which the swab would be stored dry, so that the examiner never touched the tube, and the swab's head only ever contacted the participant's conjunctiva.

In The Gambia, both samples were collected with the POC test's sterile polyurethane swab (Becton, Dickinson and Company, Franklin Lakes, USA). In Senegal, the first sample was collected with the POC test's swab, and the second sample was with a dry Dacron polyester-tipped swab (Quelab Laboratories, Montreal, Canada). This swab change was because inhibition in Studies 1 and 2 was believed to be due to the polyurethane swab, as a cloudy lysate was observed in the Amplicor extract.



Figure 1. Flowchart outlining study methods and results. doi:10.1371/journal.pntd.0001234.g001

The first swab was processed immediately in the field by the POC test. The second-collected swabs, to later be tested for the detection of ocular *C. trachomatis* with the qualitative PCR Amplicor *Chlamydia trachomatis*/*Neisseria gonorrhoeae* (CT/NG) Test (Roche Molecular Systems, Indianapolis, IN, USA), were stored in a cool box in the field and archived frozen at -20° C within ten hours of collection.

POC testing

POC testing was carried out according to the POC test's protocol [9]. Briefly, the eye swab was placed in a sample preparation tube to which three reagents were added for the release of *Chlamydia* LPS. Five drops of the sample extract were transferred to a detection tube, rehydrating two lyophilised signal amplification reagents. A dipstick was then placed inside the tube and the mixture was left to wick up for 25 minutes before the results were read.

The same person performed all POC testing and was masked to the clinical diagnosis. Results were read at 25 minutes by two different readers each masked to the other's grading. The first reader was trained by the Diagnostics Development Unit, and the second reader was trained by the first reader. Grading was practised on non-clinical samples prior to participant sample collection. The signal strength was graded from 0 (negative) to 5 (strongly positive) using a signal grading card with increments of 0.5. A positive sample is defined as any signal with a signal strength of 0.5 or more noted by the reader. In Senegal, a pocket size temperature/humidity handheld datalogger (RH32 Series, Omega, Manchester, UK) was used with values measured every 30 minutes.

Amplicor PCR processing

Amplicor, which detects the multi-copy cryptic plasmid, was performed on the second-collected swab. Amplicor was chosen as the reference test due to its good diagnostic performance on ocular samples [16,17,18], its history of use for detection of ocular *C. trachomatis* detection [15,19,20,21], and its use as the reference test in the previous evaluation of this prototype POC test [9].

Study 1 samples were tested within 42 days of collection at the Medical Research Council (MRC) Laboratories, Fajara, The Gambia. About half of Study 2 samples were processed within 1 month at the MRC, and the remainder within 4 months at the London School of Hygiene & Tropical Medicine (LSHTM). All Senegalese samples were processed at the LSHTM between 2 and 6 months of collection. A previously published [15] sample preparation protocol was used instead of that in the Amplicor package insert. Positive and negative controls provided with the assay were included to validate the runs. When clusters of positives were observed on the detection plate, the positive samples were retested on-site. Those confirmed positive on the retest were considered Amplicor positives, and the others were considered negatives. Amplification of both the plasmid DNA and the master-mix internal control sequence was tested, allowing for inhibition to be detected. Inhibited samples were diluted from 1/5 up to 1/100 with a 50:50 lysis:diluent mix, until inhibition was resolved.

Quantitative PCR processing

The bacterial load of Amplicor positive samples was estimated by processing the samples with a real-time quantitative PCR assay targeting the single-copy *ompA* gene [15]. The reverse primer, common to all ocular serovars, was 5'-TTT AGG TTT AGA TTG AGC ATA TTG GA-3'. The serovar A and B forward primers were 5'-GCT GTG GTT GAG CTT TAT ACA GAC AC-3' and 5'-TCT GTT GTT GAG TTG TAT ACA GAT AC-3' (Sigma-Genosys, Gillingham, UK), respectively. Quantitation was done on two 4 μ L replicate samples for both serovar A and serovar B. The Gambian samples were processed in a LightCycler (Roche Diagnostics, Indianapolis, USA). The Senegalese samples were processed on a Rotor-Gene RG3000 (Qiagen, Crawley, UK).

Quality control

Protocol changes were introduced as the study progressed to help ensure data quality. These changes involved the introduction of a POC test panel to be performed in the field, mock swabs inserted between patient samples in the field, environmental controls (air, loupe and glove swabs), and testing for laboratory contamination.

Panel. For Studies 2 and 3, a panel of positive (low, medium and high concentrations of *C. trachomatis* LGV-L1) and negative controls was processed at the beginning of each working morning and afternoon to check that the POC test was working correctly. A swab was dipped into the control vial (for the negative control: 10 mM PBS, 0.1% sodium azide and 1% treated casein; concentrations for the positive controls which should all be POC test positive: High: 1.143×10^7 *C. trachomatis* elementary bodies (EB)/ml; Medium: 9×10^5 EB/ml; Low: 2.25×10^5 EB/ml) and processed normally. Before the addition of the third reagent, 50 µL of these controls was aliquoted into 200 µL of pre-dispensed Amplicor diluent to be later processed by Amplicor, to serve as negative and positive controls from the field to the laboratory. These aliquoted panels were stored in the same way as the ocular swabs.

In Senegal, two types of field sample control were introduced: **Environmental controls.** At approximately every hundredth sample, a swab was waved in the air (air control), another swab wiped the top of the glove box (glove control), and a third swab wiped the front of the loupe (loupe control).

Mock swabs. Pre-prepared mock swabs were introduced between patient samples in the field before POC testing. Positive swabs were prepared with a non-ocular *C. trachomatis* strain (LGV-L1). The low load positive was below the POC test detection limit (2500 chlamydial EBs per test [9]) and the high load positive was above the detection limit. Negative mock swabs were also included. Mock swabs were introduced at a frequency which meant that one of each type of specimen would be processed per 89 patient samples. The samples were labelled in the same way as normal swabs to mask the POC and Amplicor test processors.

Amplicor quality control. At both LSHTM and MRC, Amplicor was conducted by experienced laboratory staff who had successfully completed a masked panel of samples. The Amplicor processors were masked to the clinical diagnoses and POC test results. As a measure of Amplicor reproducibility, the Amplicorpositive samples from Study 2 (20 tested at MRC, 15 tested at LSHTM) were re-tested by Amplicor at the University of Cambridge, and 21 Amplicor-negative samples were re-tested at LSHTM by a scientific officer who did not know the samples' origin. As part of the ongoing improvement to quality assurance throughout the study, it was decided to check for laboratory contamination at the time of processing the Senegalese samples. Swabs were taken of the laboratory cabinet surfaces and gloves, and processed by Amplicor.

To verify the quality of sample collection, Study 1 samples were tested for the presence of human-specific hypervariable 1 (HV1) D-loop region mitochondrial DNA (mtDNA) [13]. mtDNA-positive results indicate that human DNA is present in the sample.

Statistical analyses

Results were double-entered by different entry clerks and verified in Microsoft Access (MS Access v2000/2003XP). Any discrepancies after verification were checked against the original paper forms. Data cleaning was performed in Stata (v9.2, STATA Corp., College Station, TX, USA).

Data analysis was performed in Stata, except for the humidity and temperature analyses which were performed in R (v 2.9.0, R Foundation for Statistical Computing, Vienna, Austria). As a result of the change in Amplicor swab type between Studies 2 and 3, and that the graders in The Gambia and Senegal were different, results from the 3 studies have not been combined.

The kappa statistic was used to assess between-grader agreement for the POC test and to assess Amplicor reproducibility. The performance (sensitivity, specificity, Positive Predictive Value (PPV) and Negative Predictive Value (NPV)) of the POC test was compared against Amplicor as the gold standard. Binomial exact 95% confidence intervals (CI) were calculated to quantify uncertainty. Proportions were compared using Pearson's chi-squared statistic. Cuzick's trend test was used to look at the relationship between quantitative load, clinical sign status and POC test result.

The effect of temperature and humidity on the POC test's performance was measured using logistic regression. A scatter plot of false-positives (FPs) and true-negatives (TNs) by temperature and humidity was made, with contours of the relative risk of FPs relative to TNs. For each TN, a bivariate Normal density function was centred on the corresponding point. At any point on the graph, a density for TNs was calculated by summing these individual densities. A similar procedure was applied to the FP results. At any point, the relative risk is the ratio of these two densities. Contours of this relative risk were then added to the scatter plot (Figure 2).



Figure 2. Relative risk of false-positives relative to truenegatives by temperature and relative humidity for Study 3. doi:10.1371/journal.pntd.0001234.g002

Results

A total of 3734 children were screened and tested. Study 1 (LRR, The Gambia) included 950 children under 10 years, Study 2 (6 villages, The Gambia) 1171, and Study 3 (Senegal) 1613. This represents participation of 88.3% (950/1076), 90.8% (1171/1289), and 96.6% (1613/1669) based on the censused population in Studies 1, 2, and 3, respectively.

Laboratory controls

During Amplicor processing of the Senegalese samples, 12 lab controls were taken to check for lab contamination (3 for the hood, 2 of the glove box, 2 of the cabinet, and 5 of the gloves). All were Amplicor-negative.

Mock swabs

In Senegal, 14 negative, 15 low load, and 11 high load mock swabs were introduced in between patient samples. Amplicor correctly detected all results. For the POC test, the number of correctly identified negative, low and high load positive cases differed significantly for reader 1 (p=0.007) and reader 2 (p = 0.004). For both readers, the POC test correctly detected the high load positives in 100% of cases. These had a signal strength ranging from 1.0 to 2.0 for reader 1, and from 1.0 to 2.5 for reader 2. For low load positives (that tested POC negative under standard laboratory conditions), 5 samples were graded as positive by both the first and second readers. An additional sample was graded positive by reader 1, and 3 other samples as positive by reader 2. Thus, a total of 9/15 low load positives were detected by the POC test in the field. The signal strength of these falsepositives ranged from 0.5 to 1.5 for both readers. For the negative controls, the readers both graded 5 samples as positive, and reader 2 additionally graded 4 samples as positive. These false-positives had signal strengths of 0.5 or 1.0 for both readers.

Environmental controls

In Senegal, there were 16 air controls, 16 glove controls, and 17 loupe controls. All were Amplicor negative. Less than half the POC test results were negative for both reader 1 (42.9%) and reader 2 (46.9%).

POC test panel

In total, there were 101 panel positive and negative controls aliquoted in the field and tested by Amplicor (52 in Study 2 and 49 in Study 3). All 63 positive controls were correctly detected by Amplicor. Of the 38 negative controls, one from The Gambia tested positive repeatedly and two initially tested equivocal but were negative when repeat tested in duplicate. There was an additional equivocal negative control result by Amplicor, but the sample was erroneously labelled only as "negative control" on the template, without specifying which negative control this was, meaning it could not be retested.

A total of 56 panels were tested by the POC test (14 in Study 2 and 42 in Study 3). All positive panels, regardless of concentration, were positive by the POC test for both readers. The proportion of all negative panels correctly recorded as negative by the POC test was 60.7% for reader 1 (85.7% for Study 2, 52.4% for Study 3), and 66.1% for reader 2 (92.9% for Study 2 and 57.1% for Study 3).

Sample quality

Of 942 Amplicor-negative samples for which sample was available in Study 1, positive results for human-specific hypervariable D-loop region mtDNA were obtained in 937 (99.5%) samples. The five mtDNA-negative samples and five samples that could not be tested for mtDNA because of insufficient material from Study 1 have been removed from analyses. Three field air controls were randomly selected and also tested for *C. trachomatis* and human mtDNA, and provided negative results.

Inhibition by swab type

Inhibition in Studies 1 and 2 which used the POC test polyurethane swab was 23.4% (220/940) and 22.8% (2671171), respectively. The proportion of inhibited samples in Study 3 was 18.2% (293/1613), so the change of swab did not make a noticeable difference to the level of inhibition. Only one inhibited sample, from Study 2, retested as Amplicor positive.

Amplicor reproducibility

Of the 35 Amplicor-positive samples from Study 2 retested at the University of Cambridge, 27 were confirmed positive (23 as positive and 4 as equivocal), 3 were negative but failed the Internal Control (IC), and 5 were negative and passed the IC. All 21 Amplicor-negatives retested as negative. These retests resulted in a kappa score between the initial and retest results of 0.73, demonstrating substantial agreement.

All 35 samples originally tested as positive were considered true positives for the analyses presented. Of the 5 negatives, three were positive by quantitative PCR with estimated loads of 5, 7 and 25 *ompA* copies/swab. The remaining two positives were isolated among a string of negatives in the field, and were not near positive samples on the Amplicor detection plate.

If a true positive was considered to be one that was positive at both LSHTM and the University of Cambridge (27 samples retested positive or equivocal), the specificity and PPV estimates remain the same. The NPV increases slightly to 98.7% (97.9-99.3) for reader 1 but decreases to 98.8% (98.0-99.4) for reader 2. The sensitivity increases for both readers, but insignificantly: 48.1% (28.7-68.1, p=0.384) for reader 1 and 51.9% (31.9-71.3, p=0.352) for reader 2.

Repeat of Amplicor positive samples clustered on the detection plate

For Study 1, none of the 3 Amplicor positives were retested. For Study 2, 10/39 Amplicor positives were retested, and 6 retested negative. For Study 3, 13/51 Amplicor positives were retested and all retested positive.

Diagnostic performance

The prevalence of active trachoma and Amplicor positives was, respectively, 11.9% and 0.3% in Study 1, 23.9% and 3.0% in Study 2, and 14.9% and 1.8% in Study 3. During field processing of the POC test, mistakes were made for 4 samples in Study 1, and 3 samples in Study 3. These samples have been removed from analyses involving the POC test. The POC test's sensitivity, specificity, PPV and NPV against Amplicor showed similar point estimates and 95% CI for the two readers (Table 1). Overall, sensitivity and PPV were low, with respective estimates ranging from 33.3%–67.9%, and 4.3%–21.0%. The specificity met the minimum target of 98% in Study 1, but not in Studies 2 or 3.

There is no evidence of a significant difference between the point estimates and corresponding 95% CI for NPV, PPV, or sensitivity between the three studies. Precision for the sensitivity estimates was low due to small numbers (Table 1). Compared with Study 1, the specificity of the POC test was significantly lower in both Study 2 (p<0.001) and Study 3 (p=0.001). In Studies 2 and 3, the specificity upper confidence bounds did not exceed 96.8%.

Table 1. Diagnostic accuracy of the POC test compared with the gold standard (Amplicor).

	PCR	POC test - reader 1			POC test - reader 2		
		Negative	Positive	Total	Negative	Positive	Total
Study 1	Negative	927	6	933	911	22	933
	Positive	2	1	3	2	1	3
	Total	929	7	936	913	23	936
		Negative	Positive	Total	Negative	Positive	Total
Study 2	Negative	1087	49	1136	1076	60	1136
	Positive	22	13	35	21	14	35
	Total	1109	62	1171	1097	74	1171
		Negative	Positive	Total	Negative	Positive	Total
Study 3	Negative	1461	121	1582	1492	90	1582
	Positive	9	19	28	11	17	28
	Total	1470	140	1610	1503	107	1610

doi:10.1371/journal.pntd.0001234.t001

Effect of relative humidity and temperature on diagnostic performance

Temperature and relative humidity data were collected for all samples from Study 3 (1584 Amplicor-negatives and 29 Amplicorpositives). Figure 2 shows contours of the relative risk (RR) for FPs relative to TNs, with shading from green to red as the RR increases (see under Statistical Analyses in the Methods section). It is apparent that the false-positive RR began to increase at temperatures above 30°C and at relative humidities below 10%. The RR of a FP is approximately three times that of a truenegative at a temperature of about 36°C and at a relative humidity of 10%, and increases more rapidly as temperature rises and humidity falls.

Plots of FP rates against temperature and relative humidity indicated an increase at temperatures above 31.4° C, and a relative humidity below 11.4° . Estimates of diagnostic accuracy calculated for samples processed above and below the 31.4° C temperature threshold showed that the specificity was significantly lower in samples processed above the threshold than below, for both the first and second POC test readers (p<0.001) (Table 2). For humidity, the specificity was significantly lower in samples processed below a threshold of 11.4% compared with those above the threshold, for both readers (p<0.001) (Table 2).

Analytic sensitivity of the POC test

Of the 67 Amplicor-positives, positive *ompA* results were obtained in 58 (86.6%) samples by quantitative PCR. The estimated number of *ompA* copies/swab ranged from 5 to 3,008,063, with a median of 670. Although a few low load PCR positives were POC test positive, the POC test consistently detected positives from 1000 *ompA* copies/swab. Although the POC test is a qualitative assay, the signal strength was scored on a scale from 0.5 (weak) to 5.0 (strong) in the field. There was a significant association between increased organism load and increased POC test signal strength (p<0.001).

Inter-grader agreement

The kappa score for inter-grader variability between the two POC test readers was lowest for Study 1 and highest for Study 3. For exact signal strength the kappa score ranged from 0.41 to 0.59, showing moderate agreement. When the results were categorised as positive (signal strength ≥ 0.5) or negative (signal strength < 0.5), the scores ranged from 0.26 to 0.68, demonstrating fair to substantial agreement.

Discussion

In this study we conducted an evaluation of a prototype POC test for the detection of ocular C. trachomatis in children aged under 10 years in The Gambia and Senegal. After following standardised field and laboratory protocols, ensuring quality assurance and data validity, the results demonstrated that in its present format, this POC test is not suitable for use in the field. Under laboratory conditions, the negative and low positive mock swabs resulted in negative POC tests. In the field, the POC test gave false-positive results for approximately half of these mock swabs. This demonstrates that the POC test does not pass quality control procedures when tested in the field. When tested on children's ocular swabs, specificity in Study 1 was excellent (99.0% and 97.6% for readers 1 and 2, respectively). This is consistent with the specificity reported from the previous evaluation of this test performed in Tanzania, where the overall specificity was 99.4% (95%CI 98.8–100) [9]. However, in Studies 2 and 3, the specificity ranged from 92.4% to 95.7%, falling short of the 98% minimum specificity required for this test.

The temperature and relative humidity data provide the most likely explanation for the lower POC test specificity in Studies 2 and 3. Study 1 was conducted in January and February, when The Gambia is experiencing its cool season. Study 2 took place just before the rainy season, when temperatures rise. In Study 3, high temperatures and low relative humidities were recorded whilst performing the test, and these conditions were shown to significantly affect the false positive rate of the POC test. These data indicate that the prototype POC test's format is not appropriate for these environmental conditions.

Evaluations of rapid POC tests for other infectious diseases have observed a detrimental effect of high temperature and humidity during test storage on performance [22,23,24]. However, we observed an effect on the test's performance in the field during processing. A review of malaria rapid diagnostic tests which also use lateral flow technology, notes that humidity and wind rapidly degrade nitrocellulose capillary flow action. This effect on reagent Table 2. Effect of temperature and humidity on performance of the POC test compared to Amplicor PCR.

Temperature threshold: 31.4°C						
	POC test – reader 1		POC test - reader 2			
	Below threshold	Above threshold	Below threshold	Above threshold		
True positive	10	9	10	7		
True negative	956	505	961	531		
False positive	21	100	16	74		
False negative	7	2	7	4		
Sensitivity	58.8 (32.9–81.6)	81.8 (48.2–97.7)	58.8 (32.9–81.6)	63.6 (30.8–89.1)		
Specificity	97.9 (96.7–98.7)	83.5 (80.3–86.4)	98.4 (97.4–99.1)	87.8 (84.9–90.3)		
PPV	32.3 (16.7–51.4)	8.3 (3.8–15.1)	38.5 (20.2–59.4)	8.6 (3.5–17.0)		
NPV	99.3 (98.5–99.7)	99.6 (98.6–100)	99.3 (98.5–99.7)	99.3 (98.1–99.8)		

Relative humidity threshold: 11.4%

	POC test - reader 1		POC test – reader 2	
	Below threshold	Above threshold	Below threshold	Above threshold
True positive	6	13	5	12
True negative	381	1080	403	1089
False positive	90	31	68	22
False negative	2	7	3	8
Sensitivity	75 (34.9–96.8)	65 (40.8–84.6)	62.5 (24.5–91.5)	60 (36.1–80.9)
Specificity	80.9 (77.0–84.3)	97.2 (96.1–98.1)	85.6 (82.1–88.6)	98 (97.0–98.8)
PPV	6.3 (2.3–13.1)	30.2 (17.2–46.1)	6.8 (2.3–15.3)	35.3 (19.7–53.5)
NPV	99.5 (98.1–99.9)	99.4 (98.7–99.7)	99.3 (97.9–99.8)	99.3 (98.6–99.7)

Values in parentheses are 95% Cl.

doi:10.1371/journal.pntd.0001234.t002

flow could result in false-positives. In addition, temperature and time could be detrimental to the test's sensitivity, as they have been reported to deconjugate the signal line antibody-indicator complex, detach the capture antibody from the nitrocellulose strip, and unfold the binding sites of antibodies [25]. Since the dipstick of the ocular C. trachomatis POC test under evaluation is made up of a nitrocellulose membrane transversely lined with mAb against chlamydial LPS and antibiotin, these are plausible explanations for the observed deleterious effect of high temperature and low relative humidity on the test's performance. These results suggest that the environmental conditions during Studies 2 and 3 were harsher than those experienced in Study 1 and Tanzania, and emphasise the importance of conducting POC test evaluations in different settings. A change in the format of the prototype POC test that prevents its performance from being affected by the dry, hot, and dusty environments in which trachoma is predominantly found [26], would no doubt improve the usefulness of this test for trachoma control.

False-positives may also have appeared as a result of the POC test's target being the genus-specific chlamydial LPS. We do not believe, however, that cross-reaction with non-*C. trachomatis* bacteria was the cause of the POC test false-positives observed in this study. As noted by Michel *et al.*, the POC test's specificity

has been established against a panel of microorganisms commonly associated with the human eye and skin (such as *Staphylococcus*, *Pseudomonas*, *Streptococcus*, *Escherichia*, *Proteus*, and *Candida*, obtained from ATCC) [9]. In addition, if cross-reaction were taking place, it would not explain the observed association between FPs with temperature and humidity.

The advantage of testing the prototype POC test in low prevalence settings was the ability to gain a good estimate of specificity. The disadvantage is that we have been unable to determine an accurate estimate of the test's sensitivity. In addition, the active disease found in this study was mild with only 6.5% of clinically active children having TI. Infection load is correlated with disease severity [4,15,27]. The consequence of lower infection loads is a lower test sensitivity, especially in an assay that detects a surface antigen as opposed to one using PCR technology. Indeed, the Tanzanian evaluation observed a lower (albeit non-significant) sensitivity (76.9%) of the POC test in the lower prevalence site (TF prevalence 12.5%) compared with a sensitivity of 85.5% where the TF prevalence was 31.5% [9]. Michel et al. (2006) noted that the assay has an analytical sensitivity of 2500 chlamydial EBs per test [9]. Our quantification demonstrated consistent detection from approximately 1000 ompA copies/swab.

In terms of the limitations of this study, there was a delay of up to 6 months between sample collection and sample processing, which could have resulted in low load positives testing negative. This may have contributed to the number of POC test false-positives observed. However, since samples were stored at -20° C, we do not believe that the DNA would have degraded and that waiting would have led to a decrease in the number of true positives.

The POC test was performed on the first-collected swab whereas the "gold standard" Amplicor testing was on the secondswab. There may be differences between the two swabs, for example, in cases where there are few EBs in the conjunctiva the first swab may not leave any for the second swab. One of the swabs may also collect more PCR-inhibiting material, such as mucous, resulting in inhibition in one of the assays. Furthermore, one swab may be passed more forcefully over the conjunctiva, collecting more DNA or inhibiting material. Michel et al. (2006) demonstrated that first-collected swabs had higher loads than second-collected swabs by comparing organism load in the firstand second-collected swabs from 13 Amplicor positive individuals. The first swab's mean EB count was 643,424 compared with 181,310 for the second swab. This should not affect the Amplicor prevalence as its detection level is in the range of 1-10 EBs [28,29]. Furthermore, Amplicor result concordance between firstand second-collected swabs has been shown to be excellent [5, 30]

There was a change in swab type between Study 2 and Study 3 because it was believed that the polyurethane swab led to inhibition. However, the swab change did not make a noticeable difference to the level of inhibition. The disadvantage of inhibition is the need to dilute the sample, which would reduce the copy number in any sample tested, resulting in Amplicor false-negatives. Since load of infection in the study sites was often low (with 37.3% of all Amplicor positives having a load of <10 ompA copies/swab or being negative), this is a distinct possibility, and could have contributed to the low specificity of the POC test.

Another possible limitation is our choice of gold standard. In the absence of a universally accepted gold standard for C. trachomatis, we chose Amplicor as it was used in the previous evaluation of this POC test [9], and it has been used in multiple studies of ocular C. trachomatis infection. Controls included to assure the quality of our gold standard produced excellent results. Air, loupe, glove and spiked mock swab field controls were all correctly identified. The Amplicor results for aliquots from the POC test control panel were correct except for one negative panel from Study 2, which was repeatedly positive, and one equivocal which could not be repeat tested because the sample name was not correctly written on the Amplicor plate template. This suggests contamination of the negative panel from the positives when aliquoting in the field, which is possible as stringent laboratory conditions cannot be maintained in such an environment. Furthermore, it was a requisite for a successful run that the Amplicor-provided positive and negative controls processed for each plate produce the correct result, indicating that contamination in the lab is unlikely. This is supported by the Amplicor negative results of swabs taken of lab surfaces to check for lab contamination. When positives clustered on the detection

References

 Resnikoff S, Pascolini D, Etya'ale D, Kocur I, Pararajasegaram R, et al. (2004) Global data on visual impairment in the year 2002. Bull World Health Organ 82: 844–851. plate were repeat tested, 6/10 retested samples from Study 2 retested negative. This could indicate that there was contamination between the wells on the detection plate, and for this reason they were considered negative in analyses. Alternatively, these samples could have been low load positives that did not repeat test positive. Of the 35 Amplicor positives retested by Amplicor at the University of Cambridge, five tested negative.. The failure to retest these five samples as positive was not unexpected as reproducibility when retesting the original sample with the same test is known to be poor for low load samples [31,32,33,34,35,36]. However, when samples that were not repeated positive at the University of Cambridge were removed from the analyses, there was no significant effect on the prototype POC test's performance.

The development of effective diagnostic tools is considered a priority for Neglected Tropical Diseases (NTDs) [37], and it is therefore important to be aware of the impact the environment can have on the operational performance of POC tests. A lateral flow platform in an open system appears not to be suitable for the environments in which NTDs, such as trachoma, are often found. A rapid, accurate, simple, and affordable POC test which can be performed in the field could be a great asset to trachoma control, particularly in low prevalence settings. The specificity of the test must be high (>98%) to prevent communities from being unnecessarily mass treated. The specificity of the prototype POC test evaluated in this study decreased as the temperature increased and relative humidity decreased, indicating the importance of field testing POC tests in the different environments in which the target disease is found, in addition to being evaluated in different prevalence settings. Until a suitable test is made available, trachoma control decisions in the field remain reliant on clinical diagnosis, potentially wasting scarce resources.

Supporting Information

Supporting Information S1 STARD checklist. (DOC)

Acknowledgments

We would like to thank the village leaders and villagers for participating in this study. Many thanks to the Gambian and Senegalese field team for invaluable help in the field. Thanks to Paul Snell for database support and Rosanna Peeling for providing the masked panels for laboratory technician validation and essential contributions to discussions about the study. Thanks to Dr. Helen Lee and Dr. Claude-Edouard Michel for providing the test, field controls, retesting of samples, and critical feedback.

Author Contributions

Conceived and designed the experiments: EHE MH JFS AS BS RB DM. Performed the experiments: EHE MH SM IS CR AA. Analyzed the data: EHE EH TE MH JFS AS BS CR RB DM. Contributed reagents/ materials/analysis tools: EHE MH CR. Wrote the paper: EHE RB DM. Acquired field data: EHE JFS AS BS MH RB. Critically read and contributed to the manuscript, saw and approved the final version: EHE MH JFS SM IS AA CR AS BS EH TE RB DM.

 Thylefors B, Dawson CR, Jones BR, West SK, Taylor HR (1987) A simple system for the assessment of trachoma and its complications. Bull World Health Organ 65: 477–483.

- Baral K, Osaki S, Shreshta B, Panta CR, Boulter A, et al. (1999) Reliability of clinical diagnosis in identifying infectious trachoma in a low-prevalence area of Nepal. Bull World Health Organ 77: 461–466.
- Burton MJ, Holland MJ, Faal N, Aryee EA, Alexander ND, et al. (2003) Which members of a community need antibiotics to control trachoma? Conjunctival Chlamydia trachomatis infection load in Gambian villages. Invest Ophthalmol Vis Sci 44: 4215–4222.
- Miller K, Schmidt G, Melese M, Alemayehu W, Yi E, et al. (2004) How reliable is the clinical exam in detecting ocular chlamydial infection? Ophthalmic Epidemiol 11: 255–262.
- Solomon AW, Harding-Esch E, Alexander ND, Aguirre A, Holland MJ, et al. (2008) Two doses of azithromycin to eliminate trachoma in a Tanzanian community. N Engl J Med 358: 1870–1871.
- Harding-Esch EM, Edwards T, Sillah A, Sarr-Sissoho I, Aryee EA, et al. (2008) Risk factors for active trachoma in The Gambia. Trans R Soc Trop Med Hyg 102: 1255–1262.
- WHO (2004) Report of the eighth meeting of the WHO Alliance for the Global Elimination of Blinding Trachoma, Geneva: Switzerland. Available: http:// www.who.int/blindness/publications/GET_8_Report.pdf.
- Michel CE, Solomon AW, Magbanua JP, Massae PA, Huang L, et al. (2006) Field evaluation of a rapid point-of-care assay for targeting antibiotic treatment for trachoma control: a comparative study. Lancet 367: 1585–1590.
- Saison F, Mahilum-Tapay L, Michel CE, Buttress ND, Nadala EC, Jr., et al. (2007) Prevalence of Chlamydia trachomatis infection among low- and high-risk Filipino women and performance of Chlamydia rapid tests in resource-limited settings. J Clin Microbiol 45: 4011–4017.
- Mahilum-Tapay L, Laitila V, Wawrzyniak JJ, Lee HH, Alexander S, et al. (2007) New point of care Chlamydia Rapid Test–bridging the gap between diagnosis and treatment: performance evaluation study. BMJ 335: 1190–1194.
- Bossuyt PM, Reitsma JB, Bruns DE, Gatsonis CA, Glasziou PP, et al. (2003) The STARD statement for reporting studies of diagnostic accuracy: explanation and elaboration. Clin Chem 49: 7–18.
- Harding-Esch EM, Edwards T, Sillah A, Sarr I, Roberts CH, et al. (2009) Active trachoma and ocular Chlamydia trachomatis infection in two Gambian regions: on course for elimination by 2020? PLoS Negl Trop Dis 3: e573.
- Landis JR, Koch GG (1977) The measurement of observer agreement for categorical data. Biometrics 33: 159–174.
- Solomon AW, Holland MJ, Burton MJ, West SK, Alexander ND, et al. (2003) Strategies for control of trachoma: observational study with quantitative PCR. Lancet 362: 198–204.
- Kessler HH, Pierer K, Stuenzner D, Auer-Grumbach P, Haller EM, et al. (1994) Rapid detection of Chlamydia trachomatis in conjunctival, pharyngeal, and urethral specimens with a new polymerase chain reaction assay. Sex Transm Dis 21: 191–195.
- Kowalski RP, Uhrin M, Karenchak LM, Sweet RL, Gordon YJ (1995) Evaluation of the polymerase chain reaction test for detecting chlamydial DNA in adult chlamydial conjunctivitis. Ophthalmology 102: 1016–1019.
- Hammerschlag MR, Roblin PM, Gelling M, Tsumura N, Jule JE, et al. (1997) Use of polymerase chain reaction for the detection of Chlamydia trachomatis in ocular and nasopharyngeal specimens from infants with conjunctivitis. Pediatr Infect Dis J 16: 293–297.
- Gower EW, Solomon AW, Burton MJ, Aguirre A, Munoz B, et al. (2006) Chlamydial Positivity of Nasal Discharge at Baseline Is Associated with Ocular Chlamydial Positivity 2 Months following Azithromycin Treatment. Invest Ophthalmol Vis Sci 47: 4767–4771.

Performance of Trachoma Point-of-Care Test

- Goldschmidt P, Rostane H, Sow M, Goepogui A, Batellier L, et al. (2006) Detection by broad-range real-time PCR assay of Chlamydia species infecting human and animals. Br J Ophthalmol 90: 1425–1429.
- Andreasen AA, Burton MJ, Holland MJ, Polley S, Faal N, et al. (2008) Chlamydia trachomatis ompA Variants in Trachoma: What Do They Tell Us? PLoS Negl Trop Dis 2: e306.
- Chanteau S, Rahalison L, Ralafiarisoa L, Foulon J, Ratsitorahina M, et al. (2003) Development and testing of a rapid diagnostic test for bubonic and pneumonic plague. Lancet 361: 211–216.
- Garcia-Lerna JG, McNulty A, Jennings C, Huang D, Heneine W, et al. (2009) Rapid decline in the efficiency of HIV drug resistance genotyping from dried blood spots (DBS) and dried plasma spots (DPS) stored at 37 degrees C and high humidity. J Antimicrob Chemother 64: 33–36.
- Nato F, Phalipon A, Nguyen TL, Diep TT, Sansonetti P, et al. (2007) Dipstick for rapid diagnosis of Shigella flexneri 2a in stool. PLoS One 2: e361.
- Murray CK, Gasser RA, Jr., Magill AJ, Miller RS (2008) Update on rapid diagnostic testing for malaria. Clin Microbiol Rev 21: 97–110.
- Schemann JF, Laffly D, Sacko D, Zephak G, Malvy D (2007) Trichiasis and geoclimatic factors in Mali. Trans R Soc Trop Med Hyg 101: 996–1003.
- West SK, Munoz B, Mkocha H, Holland MJ, Aguirre A, et al. (2005) Infection with Chlamydia trachomatis after mass treatment of a trachoma hyperendemic community in Tanzania: a longitudinal study. Lancet 366: 1296–1300.
- Miyashita N, Lijima Y, Matsumoto A (1994) Evaluation of the sensitivity and specificity of polymerase chain reaction test kit, AMPLICOR Chlamydia trachomatis. Microbiol Immunol 38: 81–85.
- Shattock RM, Patrizio C, Simmonds P, Sutherland S (1998) Detection of Chlamydia trachomatis in genital swabs: comparison of commercial and in house amplification methods with culture. Sex Transm Infect 74: 289–293.
- Chidambaram JD, Alemayehu W, Melese M, Lakew T, Yi E, et al. (2006) Effect of a single mass antibiotic distribution on the prevalence of infectious trachoma. Jama 295: 1142–1146.
- Mulcahy GM, Albanese EA, Bachl BL (1998) Reproducibility of the Roche Amplicor polymerase chain reaction assay for detection of infection by Chlamydia trachomatis in endocervical specimens. Clin Chem 44: 1575–1578.
- 32. Castriciano S, Luinstra K, Jang D, Patel J, Mahony J, et al. (2002) Accuracy of results obtained by performing a second ligase chain reaction assay and PCR analysis on urine samples with positive or near-cutoff results in the LCx test for Chlamydia trachomatis. J Clin Microbiol 40: 2632–2634.
- Culler EE, Caliendo AM, Nolte FS (2003) Reproducibility of positive test results in the BDProbeTec ET system for detection of Chlamydia trachomatis and Neisseria gonorrhoeae. J Clin Microbiol 41: 3911–3914.
- Verkooyen RP, Noordhoek GT, Klapper PE, Reid J, Schirm J, et al. (2003) Reliability of nucleic acid amplification methods for detection of Chlamydia trachomatis in urine: results of the first international collaborative quality control study among 96 laboratories. J Clin Microbiol 41: 3013–3016.
- Schachter J, Chow JM, Howard H, Bolan G, Moncada J (2006) Detection of Chlamydia trachomatis by nucleic acid amplification testing: our evaluation suggests that CDC-recommended approaches for confirmatory testing are illadvised. J Clin Microbiol 44: 2512–2517.
- de Barbeyrac B, Goldschmidt P, Malembic S, Raherison S, Clerc M, et al. (2007) Quality assessment of conjunctival specimens for detection of Chlamydia trachomatis by PCR in children with active trachoma. Clin Microbiol Infect 13: 689–694.
- Hotez PJ, Molyneux DH, Fenwick A, Kumaresan J, Sachs SE, et al. (2007) Control of neglected tropical diseases. N Engl J Med 357: 1018–1027.